Author's response to reviews

Title: Nasal and perirectal colonization of vancomycin sensitive and resistant enterococci in patients of paediatrics ICU (PICU) of tertiary health care facilities

Authors:

Muhammad A Yameen (arfatyameen@yahoo.com)
Naeem Akhtar (naeeakh@yahoo.com)
Saira Iram (sevielle.erum@yahoo.com)
Abdul Mannan (abdulmannan_ka@yahoo.com)
Shujaat A Khan (drshujaatalikhan@ciit.net.pk)

Version: 5 Date: 27 February 2013

Author's response to reviews: see over
The BMC Infectious Diseases Editorial Team

Enclosed, please find the revised version of our manuscript entitled "Nasal and perirectal colonization of vancomycin sensitive and resistant enterococci in patients of paediatrics ICU (PICU) of tertiary health care facilities".

In your letter, you asked us to include the name of the specific ethics committee that approved the study within the methods section. We have revised the method section of manuscript accordingly.

We are again very grateful to editorial team and the reviewers for the time they spent reviewing our manuscript.

Thanks and regards

Muhammad ArfatYameen